Author’s response to reviews

Title: Normal/high-fat milk consumption is associated with higher lean body and muscle mass in Japanese women aged between 40 and 60 years: a cross-sectional study

Authors:

Yuri Sukenobe (uef1j06@gmail.com)
Masakazu Terauchi (teragyne@tmd.ac.jp)
Asuka Hirose (asuka.kajiyama@gmail.com)
Miho Hirano (120612ms@tmd.ac.jp)
Mihoko Akiyoshi (akiyoshi@shonan.bunkyo.ac.jp)
Kiyoko Kato (kiyoko2226@nifty.com)
Naoyuki Miyasaka (n.miyasaka.gyne@tmd.ac.jp)

Version: 4 Date: 24 Jan 2018

Author’s response to reviews:

Response to the Editor

We truly appreciate valuable suggestions and comments made by the editor. Below are our point by point responses.

Editor Comments:

> 1. In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data.

According to the editor’s comment, the sentence below was placed in the “Availability of data and materials” section:

“The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request.”

> 2. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship Currently, MH, MA, KK and NM are missing.
According to the editor’s comment, the sentences below were inserted into the “Author’s contributions” section:

“MH, MA, and KK were responsible for data collection. NM participated in the project development and supervision.”

> 3. Please provide figure legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Figure files should contain only the image, as well as any associated keys/annotations. If legends are present within the figure files, please remove them.

According to the editor’s comment, the title and legends of the Figure 1 were removed and placed under the new heading “Figure legends” in the main text.

> 4. Please include separate funding and competing interests sections in your declarations.

According to the editor’s comment, “Funding” and “Competing interests” were described separately.